Author's response to reviews

Title: Knowledge, Attitudes and Practices of Healthcare Ethics and Law among Doctors and Nurses in Barbados

Authors:

Seetharaman Harihan (uwi.harihan@gmail.com)
Ramesh Jonnalagadda (rameshj@caribsurf.com)
Harley Moseley (hslm2@caribsurf.com)
Errol Walrond (erwalrond@caribsurf.com)

Version: 3 Date: 1 June 2006

Author's response to reviews: see over
To
The Editorial Team
BMC Medical Ethics

Re: MS: 6156645679393208 - Knowledge, Attitudes and Practices of Healthcare Ethics and Law among Doctors and Nurses in Barbados

Many thanks for reassessing the above article and providing comments. We have accordingly revised the manuscript incorporating the suggestions by the reviewer.

The following is the list of the changes made in the manuscript:

- The methods section has been modified. However, instead of having more tables or boxes, we have attached the questionnaire as an appendix
- Language corrections have been made
- The article has been formatted in accordance with the ‘instructions for authors’ given by BioMed Central

We request that the present revised version may kindly be reviewed again and considered for publication.

Thanking You,

Sincerely,

[Signature]

Seetharaman Hariharan